Author’s response to reviews

Title: A survey of the working status of family medicine physicians in clinics and hospitals in Korea

Authors:

Hyun-Young Shin (shydeborah@gmail.com; shy801117@gmail.com)

Version: 3 Date: 27 Mar 2020

Editor Comments:
We are satisfied with the revisions made in response to the reviewer's comments, however there are a few points we would like to ask you to address:

1. Limitations
When discussing your research limitations, do not just provide the list of shortcomings of your work. It is also important for you to explain how these limitations have impacted your research findings. Therefore, please revise the limitations of your study taking into consideration the following points:
- describe all limitations of the study and explain their relevance;
- explain the nature of the limitations and justify the choices you made during the research process;
- assess the impact of each limitation to the findings and conclusions of your study;
- suggest how the limitations could be overcome in the future.

Thank you for your feedbacks. Based on the advice you gave, we revised the limitations in the discussion session. (page 9, line no. 10)

2. Language
Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed.

We got English editing service and corrected the grammar, wording, spelling, and/or punctuation.

BMC Family Practice operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.

Reviewer reports:

Imam Xierali (Reviewer 1): na.

Thank you for your feedback.